Author's response to reviews

Title: Morphologic and Immunophenotypic Evidence of In-Situ Kaposi Sarcoma

Authors:

Panagiotis A Konstantinopoulos (pkonstan@bidmc.harvard.edu)
Bruce J Dezube (bdezube@bidmc.harvard.edu)
Liron Pantanowitz (Liron.Pantanowitz@bhs.org)

Version: 3 Date: 5 October 2006

Author's response to reviews: see over
Dear Editor,

We would like to thank the reviewers for their thoughtful evaluation of our manuscript and for their most welcome comments/suggestions. Accordingly, we have now revised our manuscript to reflect these comments. All inserted changes are in bold and underlined characters. The revised manuscript conforms to your journal’s guidelines for authors. Specifically:

**Reviewer 1**

Comment 1: The article by Ramdial et al. was referenced according to the reviewer’s suggestion.

**Reviewer 2**

Comment 1: To address the reviewer’s comment, we have now included a more detailed presentation of the case incorporating all the specific clinical information requested by the reviewer.

Comment 2: To address the reviewer’s suggestion, we have now included an additional Figure (Figure 4) that illustrates the dual immunostaining.

Sincerely,

The authors,